Author's response to reviews

Title: The Aryl Hydrocarbon Receptor Ligand Omeprazole Inhibits Breast Cancer Cell Invasion and Metastasis

Authors:

Un-Ho Jin (jinunho@gmail.com)
Syng-Ook Lee (synglee@ibt.tamhsc.edu)
Catherine Pfent (cpfent@cvm.tamu.edu)
Stephen Safe (ssafe@cvm.tamu.edu)

Version: 4
Date: 1 July 2014

Author's response to reviews: see over
July 1, 2014

Ms. Patricia Ratin
c/o Prof. John Hartley
BMC Cancer

Dear Ms. Ratin:

We would like to resubmit our revised manuscript (MS # 1296835739106949) to BMC Cancer. We have made the following requested changes:

1. Email addresses for each author have been added to the Title Page.

2. The sources of all cell lines have been added to the Methods section.

3. The supplementary figure captions have been added after the current figure legends.

We hope the paper is now acceptable.

Sincerely yours,
Stephen Safe

Stephen Safe, D.Phil.
Distinguished and Regent's Professor of Toxicology
Department of Veterinary Physiology and Pharmacology
Director, Center for Environmental and Genetic Medicine
Institute of Biosciences and Technology
Texas A & M University
Tel: 979-845-5988
Fax: 979-862-4929
Email: ssafe@cvm.tamu.edu